Author’s response to reviews

Title: Anticancer activity of a sub-fraction of dichloromethane extract of Strobilanthes crispus on human breast and prostate cancer cells in vitro

Authors:

- Nik-Soriani Yaacob (soriani@kb.usm.my)
- Nurraihana Hamzah (nraihana@yahoo.com)
- Nik-Nursyazni Nik-Mohamed Kamal (geneflow143@yahoo.com)
- Siti A Zainal Abidin (amalina_sza@yahoo.com)
- Choon-Sheen Lai (choonshen@yahoo.com)
- Visweswaran Navaratnam (Nava@usm.my)
- Mohd-Nor Norazmi (norazmi@kb.usm.my)

Version: 5 Date: 3 August 2010

Author’s response to reviews:

Dr. Melissa Norton
Editor-in-Chief,
BMC Complementary and Alternative Medicine

3 August 2010

Dear Dr. Norton,

With reference to your last email regarding the formatting changes to be made to the manuscript, I hereby provide the point-by-point response to your queries:

1. Title: please ensure that the title is identical in the manuscript and the submission system. Currently they do not match.
   - Corrected as proposed

2. Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs). Currently they do not match.
   - Corrected as proposed

3. References: References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example 'text [1].' Reference 35 is not currently cited in the manuscript. Please ensure that all references are cited in numerical order once this reference has been cited.
   - Reference 35 is included and properly cited

4. Text: Please ensure that all text is displayed in black throughout the manuscript.
5. Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures.

- All figures are cropped as required

6. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

- Manuscript checked and edited as required.

Thank you

Sincerely,

Nik Soriani Yaacob
Universiti Sains Malaysia
Corresponding Author